Author's response to reviews

Title: Using intervention mapping to deconstruct cognitive work hardening: A return-to-work intervention for people with depression

Authors:

Adeena Wisenthal (adeena.wisenthal@queensu.ca)
Terry Krupa (terry.krupa@queensu.ca)

Version: 3
Date: 3 February 2014

Author's response to reviews:

MS: 5011112031197374
Research article
Using intervention mapping to deconstruct cognitive work hardening: A return-to-work intervention for people with depression
Adeena Wisenthal and Terry Krupa
BMC Health Services Research

February 2, 2014
Re: Emails sent on January 28 and January 31, 2014 from Journal Editorial Office, Biomed Central:

Thank you for the email regarding our manuscript. The following addresses the points raised.

1. Acknowledgements
This section has been added to the manuscript with a statement indicating that there are no individuals or funding organizations that contributed directly or indirectly to this manuscript.

2. Ethics statement
A statement was added to the Methods section of the manuscript indicating that an ethics statement is not included as there was no study conducted and there was no research conducted on human subjects.

Other minor changes to the original manuscript were made to conform to the journal style. A revised manuscript is being submitted at this time.

Regards,
Adeena Wisenthal, corresponding author
adeena.wisenthal@queensu.ca